Author's response to reviews

Title: Sex partnerships, health, and social risks of young men leaving jail: analyzing data from a randomized controlled trial

Authors:

  Megha Ramaswamy (mramaswamy@kumc.edu)
  Nicholas Freudenberg (nfreuden@hunter.cuny.edu)

Version: 3 Date: 1 November 2010

Author's response to reviews: see over
Editorial Board
BMC Public Health

November 1, 2010

Subject: Revision and Resubmission of MS: 1313624437933023, Sex Partnerships, Health, and Social Risks of Young Men Leaving Jail: Analyzing Data from a Randomized Controlled Trial, Megha Ramaswamy and Nicholas Freudenberg

Dear BMC Public Health Editorial Board:

Thank you so much for considering our manuscript (referenced above) for publication in *BMC Public Health*. We sincerely appreciate the valuable reviewer feedback and conditional acceptance. Please see below for clarification on a couple of items. The manuscript has also been uploaded for publication (with tables appended at end, after references).

Feedback from BioMed Central Editorial:
1. BioMed Central Editorial wanted to know if the trial referred to in this manuscript was registered: the trial was *not* registered, and therefore does not have a TRN.
2. BioMed Central Editorial would like the authors to make sure the paper is formatted per rules in the checklist: the authors checked and made small changes to the manuscript to ensure formatting was correct (e.g. removed unnecessary capitalization from titles).

Thank you very much for conditional acceptance of this manuscript in *BMC Public Health*. And please extend our sincere thanks to the reviewers for their thoughtful comments in the first round of reviews- they were very helpful. If you have any questions or concerns, please do not hesitate to contact me directly.

Best,

Megha Ramaswamy, PhD, MPH
University of Kansas School of Medicine
Kansas City, Kansas
USA
mramaswamy@kumc.edu